

Clinical Practice Guideline

Undernutrition in Chronic Kidney Disease

Final Version:	June 2019
Review Date:	June 2024

Authors:

Dr Mark Wright

Consultant Nephrologist, St James's University Hospital, Leeds

Mrs Elizabeth Southcott

Senior Specialist Renal Dietitian, St James's University Hospital, Leeds

Dr Helen MacLaughlin,

Consultant Dietitian, Kings College Hospital

Mr Stuart Wineberg

Patient Representative



Endorsements



The National Institute for Health and Care Excellence (NICE) has accredited the process used by the Renal Association to produce its Clinical Practice Guidelines. Accreditation is valid for 5 years from January 2017. More information on accreditation can be viewed at <u>www.nice.org.uk/accreditation</u>



The Renal Nutrition Group (RNG) of the British Dietetic Association acknowledge there is limited evidence for the recommendation of routine supplementation with water soluble vitamins for patients on haemodialysis. The evidence is limited to the DOPPS study (recommendation, 1c, 2m), where supplementation with water soluble vitamins showed significantly lower mortality rates (16% at patient and institution level). However, as many centres in the UK have implemented this recommendation and since there is no evidence of harm and/or toxicity, it is the RNG committee's opinion that patients on HD should be offered routine water soluble vitamins supplementation.

Method used to arrive at a recommendation

The recommendations for the second draft of this guideline resulted from a collective decision reached by informal discussion by the authors and, whenever necessary, with input from the Chair of the Clinical Practice Guidelines Committee. If no agreement had been reached on the appropriate grading of a recommendation, a vote would have been held and the majority opinion carried. However this was not necessary for this guideline.

Conflicts of Interest Statement

All authors made declarations of interest in line with the policy in the Renal Association Clinical Practice Guidelines Development Manual. Further details can be obtained on request from the Renal Association.

Acknowledgements

This document has been externally reviewed by key stake holders according to the process described in the Clinical Practice Guidelines Development Policy Manual. The authors would like to express their gratitude to the British Dietetic Association Renal Nutrition Group for their thoughts and advice during the development of this guideline.



Table of Contents

1.	Introduction4
2.	Summary of clinical practice guidelines7
3.	Summary of audit measures10
4.	Rationale for clinical practice guidelines11
5.	Lay Summary



1. Introduction

Background

"Malnutrition" describes both over and undernutrition. In the UK, the National Institute for Health and Care Excellence (NICE) define malnutrition as "a state in which a deficiency of nutrients such as energy, protein, vitamins and minerals causes measurable adverse effects on body composition, function or clinical outcome" in their Clinical Guideline (CG 32)¹ and Quality Standard (QS24)². These guidelines suggest that patients at high risk of malnutrition should be screened and referred for specialist support if it is present.

Patients with kidney failure face a number of challenges to their nutritional balance³. Uraemia itself inhibits appetite and reduces nutrient intake. Dialysis treatments result in the loss of small amounts of nutrients too. There are also other factors that tend to promote protein breakdown over protein synthesis. These include acidosis, insulin resistance and chronic inflammation. This latter factor can coexist with extensive atherosclerotic disease in a particularly toxic combination that is associated with increased mortality risk³. The International Society of Renal Nutrition and Metabolism (ISRNM), recommend the term protein energy wasting (PEW) to describe the undernutrition that is prevalent in renal failure populations⁴, though more than one underlying cause may present in any individual.

Protein energy wasting is described in 20-40% of patients with stage 4-5 chronic kidney disease $(CKD)^3$. It is a more frequent finding in dialysis patients where it is described in 28-54%⁵ and is associated with reduced survival, poor healing, infection risk, impaired functional ability and reduced quality of life³.

Aim

This clinical practice guideline reviews existing recommendations from International Society for Renal Nutrition & Metabolism, European Society for Parenteral and Enteral Nutrition, Kidney Disease Improving Global Outcomes, American Society for Parenteral and Enteral Nutrition and European Renal Association/European Dialysis & Transplant Association ⁶⁻¹⁰ and considers recent guidance on advanced kidney failure from NICE¹¹ with the aim of reducing variation in practice.

Scope

This guideline has considered how to help adults with CKD stage 4 and 5.

We have not explored the evidence relating to kidney disease in children.

This guideline is intended for health care professionals and people with kidney disease.



Appraisal of Evidence and Development of Recommendations

The modified GRADE system was used in accordance with the Renal Association's "Clinical Practice Guideline Development Manual"¹².

There is a two-level grading system for the strength of recommendations.

A **Grade 1** recommendation is a strong recommendation to do (or not do) something, where the benefits clearly outweigh the risks (or vice versa) for most, if not all patients.

A **Grade 2** recommendation is a weaker recommendation, where the risks and benefits are more closely balanced or are more uncertain.

Explicit methodology is used to describe the quality of evidence.

Grade A evidence means high-quality evidence that comes from consistent results from wellperformed randomised controlled trials, or overwhelming evidence of some other sort (such as wellexecuted observational studies with very strong effects).

Grade B evidence means moderate-quality evidence from randomised trials that suffer from serious flaws in conduct, inconsistency, indirectness, imprecise estimates, reporting bias, or some combination of these limitations, or from other study designs with special strength.

Grade C evidence means low-quality evidence from observational studies, or from controlled trials with several very serious limitations.

Grade D evidence is based only on case studies or expert opinion.

Clinical Issues covered

We considered how best to identify those at risk of undernutrition and how best to reduce this risk.

We used the HDAS database search tool that is accessible via Health Education England and NICE to identify information sources. This includes a number of search tools including Pub Med, EMBASE, CINAHL and Medline. This was most recently accessed in November 2018. Search terms included, "screening", "kidney disease", "nutrition", "malnutrition", "vitamin", and other terms pertinent to dialysis as deemed necessary. We also searched the Cochrane library for relevant systematic reviews.

References

- **1.** National Institute of Health and Care Excellence. Nutrition support for adults: oral nutrition support, enteral tube feeding and parenteral nutrition.2006; nice.org.uk/guidance/cg32
- 2. National Institute of Health and Care Excellence. Nutrition support in adults. 2012; nice.org.uk/guidance/qs24



- **3.** Lodebo BT, Shah A, Kopple JD. Is it important to prevent and treat protein-energy wasting in chronic kidney disease and chronic dialysis patients? Journal of Renal Nutrition 2018; 28:369-379
- 4. FouqueD, Kalantar-Zadeh K, Kopple J, Cano N, Chauveau P, Cuppari L, Franch H, Guarnieri G, Ikizler ATA, Kaysen G, Lindholm B, Massy Z, Mitch W, Pineda E, Stenvinkel P, Trevinho-Becerra A, Wanner C. A proposed nomenclature and diagnostic criteria for protein-energy wasting in acute and chronic disease. Kidney Int 2008; 73, 391-398
- 5. Carrero JJ,Thomas F, Nagy, Arogundade K, Avesani CM,Chan M, Chmielewski M, Cordeiro AC, Espinosa-Cuevas A, Fiaccadori E, Guebre-Egziabher F, Hand RK, Hung AM, Ikizler TA, Johansson LR, Kalantar-Zadeh K, Karupaiah T, Lindholm B, .Kovesdy CP. Global prevalence of protein-energy wasting in kidney disease: a meta-analysis of contemporary observational studies from the international society of renal nutrition and metabolism. Journal of Renal Nutrition 2018; 28; 380-392
- 6. Ikizler TA, Cano N, Franch H, Fouque D, Himmelfarb J, Kalantar-Zadeh K, Kuhlmann MK, Stenvinkel P, TerWee P, Teta D, Wang AYM, Wanner C. Prevention and treatment of protein energy wasting in chronic kidney disease patients: a consensus statement by the International Society of Renal Nutrition and Metabolism. Kidney Int 2013; 84, 1096-1107.
- Fouque D, Vennegoor M, Wee PT, Wanner C, Basci A, Canaud B, Haage P, Konner K, Kooman J, Martin-Malo A, Pedrini L, Pizzarelli F, Tattersall J, Tordoir J, and Vanholder R. EBPG Guideline on nutrition Nephrology Dialysis Transplantation 2007; 22 (supplement 2): ii22-ii44
- 8. Cano N, Fiaccadori E, Tesinsky P, Toigo G, Druml W, Kuhlmann M, Mann H, Horl WH, ESPEN guidelines on enteral nutrition: adult renal failure. Clin Nutr 2006; 25, 295-310
- KDIGO Kidney Disease: Improving Global Outcomes Working Group. KDIGO 2012 clinical practice guideline for the evaluation and management of chronic kidney disease. Kidney Int. Suppl 2013, 3, 1-150
- 10. White JV, Guenter P, Jensen G, Malone A, Schofield M. Consensus statement: Academy of Nutrition and Dietetics and American Society for Parenteral and Enteral Nutrition: Characteristics recommended for the identification and documentation of adult malnutrition (undernutrition). Journal of Parenteral and Enteral Nutrition 2012; 36, 275-283
- **11.** National Institute of Health and Care Excellence. Renal replacement therapy and conservative management. 2018; nice.org.uk/guidance/ng107
- Renal Association Clinical Guideline Committee. Clinical Practice Guideline Development Manual 2016. https://renal.org/wp-content/uploads/2017/06/renal-association-guidelinedevelopment-manual.pdf



2. Summary of clinical practice guidelines

1. Identification of undernutrition in people with kidney disease (Guidelines 1.1 – 1.3)

Guideline 1.1 – Screening for risk of undernutrition in CKD

We suggest that patients with stages 4-5 CKD should be screened to identify those at risk of undernutrition (2C):

We suggest that screening should be performed (2D);

- On admission then weekly for inpatients
- At clinic review for outpatients with stage 4-5 CKD
- 2-3 monthly for stable haemodialysis patients
- 2-3 monthly for stable peritoneal dialysis patients
- Screening should be performed earlier if their clinical condition changes

Guideline 1.2 – Diagnosis of undernutrition in people with kidney disease

1.2.1 - We suggest that there should be a clear pathway for prompt referral to specialist renal dietitians when risk of undernutrition is identified. This pathway should include locally agreed timescales for formal assessment. (2D)

1.2.2 - We suggest that patients should be assessed by a specialist renal dietitian when they begin education about renal replacement treatment and within one month of starting dialysis or changing dialysis modality (2D)

1.2.3 - We suggest that formal nutritional assessments are carried out on those identified to be at risk by screening. These diagnostic assessments will typically be performed by specialist renal dietitians with support from the broader multidisciplinary team. (2B)

Guideline 1.3 - We recommend that in-patients at risk of malnutrition on screening are also considered at risk of refeeding syndrome. (1D)



2. Interventions to reduce the prevalence of undernutrition in people with kidney disease (Guidelines 2.1 - 2.6)

Guideline 2.1 – Dose of small solute removal to prevent anorexia

We recommend that dialysis dose meets recommended solute clearance index guidelines (e.g. URR, Kt/V). (1C)

Guideline 2.2 – Correction of metabolic acidosis

We suggest that venous bicarbonate concentrations should be maintained in the normal range (2C)

Guideline 2.3 – Daily dietary protein intake

We recommend a protein intake of:

- 0.8-1.0 g/kg ideal body weight (IBW)/day for patients with stage 4-5 CKD not on dialysis (1C)
- 1.1-1.4 g/kg IBW/day for patients treated on maintenance haemodialysis (1C)
- 1.0-1.2 g/kg IBW/day for patients treated with peritoneal dialysis (1C)
- This should be accompanied by an adequate energy intake. (1C)
- We do not believe there is sufficient evidence to routinely recommend low protein diets for people with progressive kidney disease (1C)

Guideline 2.4 – Daily energy intake

We suggest a prescribed energy intake of:

- 30-40 kcal/kg IBW/day for all patients depending upon age and physical activity (2C).
- We note that peritoneal dialysis patients are likely to absorb glucose from their dialysis fluid and this should be taken into account.

Guideline 2.5 – Micronutrient supplementation in patients on dialysis

We suggest that water soluble vitamin supplements should be offered to patients on dialysis with a reduced nutrient intake or those that have unusually high levels of solute clearance on dialysis (e.g. daily or overnight haemodialysis). (2C)

We recommend that other micronutrients are supplemented only if there are symptoms consistent with deficiency and biochemical evidence of deficiency. (1C)



3. Interventions to treat undernutrition in people with kidney disease (Guidelines 3.1 - 3.3)

Guideline 3.1 Anabolic agents in kidney disease

We recommend that anabolic agents should not be used to treat undernutrition in people with kidney disease. (1C)

Guideline 3.2 – Oral nutritional supplements in patients who are undernourished

We recommend the use of oral nutritional supplements (ONS) when nutritional intake fails to increase, despite intervention and advice, and remains inadequate to meet energy and protein requirements. (1C)

Guideline 3.3 – Enteral feeding in patients who are undernourished

We suggest that the use of enteral tube feeding is considered in selected cases if nutrient intake is suboptimal despite oral nutritional support recognising that there are significant risks and inconvenience associated with these forms of feeding (2C). It is important to consider the patient's comorbidity, general condition and likely survival prospects before initiating enteral tube feeding.

Guideline 3.4 – Parenteral nutritional support in patients who are undernourished

We suggest intradialytic parenteral nutrition (IDPN) in haemodialysis or intraperitoneal amino acids in peritoneal dialysis may be considered for selected cases when oral or enteral intake is suboptimal (2D).



3. Summary of audit measures

Audit Measure 1: The service should be able to demonstrate that there is a clear pathway for nutrition care that states how patients at risk of undernutrition will be identified, who they should be referred to and timescales for formal assessment.

Audit Measure 2: The service should be able to demonstrate that "at risk" patients were assessed by a specialist renal dietitian within the locally agreed timeframe.

Audit Measure 3: The service should be able to demonstrate that all patients commencing dialysis (or changing modality) are assessed by a specialist renal dietitian within four weeks.



4. Rationale for clinical practice guidelines

1. Identification of undernutrition in people with kidney disease (Guidelines 1.1 – 1.3)

Guideline 1.1 – Screening for risk of undernutrition in CKD

We suggest that patients with stages 4-5 CKD should be screened to identify those at risk of undernutrition (2C):

We suggest that screening should be performed (2D);

- On admission then weekly for inpatients
- At clinic review for outpatients with stage 4-5 CKD
- 2-3 monthly for stable haemodialysis patients
- 2-3 monthly for stable peritoneal dialysis patients
- Screening should be performed earlier if their clinical condition changes

Rationale

Protein energy wasting is common in people with kidney disease¹. In the whole UK population, malnutrition is estimated to cost £13 billion². It is estimated to cost twice as much to treat patients who are malnourished in hospital than their well-nourished counterparts³. This has led to a policy of screening people to see if they are at risk of having malnutrition^{4,5}. Early identification means that some contributing factors can be corrected and progression may be delayed or reversed. That said, research evidence demonstrating improvements in mortality, hospital admissions or length of stay, whether it be in the general population⁶ or those with kidney disease, is not strong^{7,8}.

- In the UK, NICE produced a quality standard on nutrition support (QS24)⁵. This was reviewed in 2017 and no changes were introduced. This set out five statements regarding nutritional screening as follows: People in care settings are screened for the risk of malnutrition using a validated screening tool.
- **2.** People who are malnourished or at risk of malnutrition have a management care plan that aims to meet their nutritional requirements.
- **3.** All people who are screened for the risk of malnutrition have their screening results and nutrition support goals (if applicable) documented and communicated in writing within and between settings.
- **4.** People managing their own artificial nutrition support and/or their carers are trained to manage their nutrition delivery system and monitor their wellbeing.
- **5.** People receiving nutrition support are offered a review of the indications, route, risks, benefits and goals of nutrition support at planned intervals



This aims to involve the whole multidisciplinary team and the patient in a conversation about nutrition. Formal assessment of nutrition status is time consuming and requires specific training and expertise. The NICE guidelines suggest that screening, which requires a simple and valid screening tool, is undertaken by health care workers to identify those at risk of malnutrition so that specialist dietitians can focus on those identified as at risk by screening.

The "Malnutrition Universal Screening Tool" (MUST) is recommended by NICE for population screening⁴. This simple system can be taught to non-specialists, but it depends upon changes in weight. People with kidney disease tend to retain fluid as their kidney problem worsens and lose fluid when dialysis starts, so measures that depend upon actual body weight are not as reliable as usual. MUST has poor sensitivity in dialysis patients^{9,10}.

The 2018 NICE guidance on renal replacement therapy (NG107) recognises that the relative number of specialist renal dietitians varies between centres⁸. They recommend assessment when patients commence dialysis or choose to pursue a conservative treatment path. They make no further recommendations about undernutrition because they felt that the available evidence lacked strength. This leaves kidney services with apparently conflicting advice.

There has been a lot research into undernutrition in kidney disease. Very few intervention studies of appreciable size look at end points like quality of life, frequency of hospital admission, length of stay in hospital or mortality. This means that it is not possible to confidently predict benefit in these parameters. It does not rule out the possibility of benefit though.

We continue to favour screening for risk of undernutrition in keeping with NICE CG32. There are a number of modified tools described in the literature. Screening tools need to be simple to use and easy to teach to non-specialist staff. They also need to be reproducible and reliable in their ability to find people at risk of having undernutrion⁵. Training materials are useful too. The work by Yamada¹⁰ Campbell¹¹, Xia¹², Jackson¹³ or MacLaughlin¹⁴ may offer a solution for your unit and we recommend that your multidisciplinary team look at these and others to decide how best to screen your patients.

There is little evidence to guide the frequency of screening. NICE recommend that outpatients should be screened at first consultation and in-patients upon admission⁴. They recommend that inpatients should be re-screened weekly⁴. Outpatients with stage 4 or 5 CKD are at high risk for undernutrition and should be screened at each outpatient visit, taking account of changes in fluid balance and recent changes in the amount of food being eaten. People with progressive stage 4 CKD are likely to benefit from education from specialist renal dietitians in terms of advice about energy, salt, potassium and phosphate intake even if they do not flag as a concern on nutritional screening tests¹⁵. We suggest that established patients on dialysis should be screened every 2-3 months. If concerns arise due to intercurrent illness, screening should be repeated at that time.



Guideline 1.2 – Diagnosis of undernutrition in people with kidney disease

1.2.1 - We suggest that there should be a clear pathway for prompt referral to specialist renal dietitians when risk of undernutrition is identified. This pathway should include locally agreed timescales for formal assessment. (2D).

1.2.2 - We suggest that patients should be assessed by a specialist renal dietitian when they begin education about renal replacement treatment and within one month of starting dialysis or changing dialysis modality (2D).

1.2.3 - We suggest that formal nutritional assessments are carried out on those identified to be at risk by screening. These diagnostic assessments will typically be performed by specialist renal dietitians with support from the broader multidisciplinary team. (2B).

Rationale

Our recommendations aim to ensure that services have clear pathways for referral to specialist renal dietitians. These individuals should have received specialist training in the techniques used to diagnose undernutrition in people with kidney disease.

People deemed to be at risk of undernutrition should be referred for assessment. This uses more sophisticated techniques to determine whether or not undernutrition is present and how severe it is and decide to guide a treatment plan.

ASPEN use the following six evidence based criteria for nutritional assessment:

- Insufficient energy intake
- Weight loss
- Loss of muscle mass
- Loss of subcutaneous fat
- Fluid accumulation that can sometimes mask weight loss
- Diminished "functional status" as measured by hand grip strength.

The presence of two or more of these would lead to a diagnosis of undernutrition¹⁶.

SGA includes gastrointestinal symptoms (appetite, anorexia, nausea, vomiting, diarrhoea), weight change in the preceding 6 months and last 2 weeks, evidence of functional impairment and a subjective visual assessment of subcutaneous tissue and muscle mass¹⁷.

Modern bioimpedance analysis or spectroscopy devices are primarily designed to assess fluid status but have also been used in nutritional assessment. They require training and expertise, but a wide variety of staff can be taught¹⁸.

Assessment should be carried out by a specialist renal dietitian with knowledge of renal disease and sufficient experience to know which of these tools will be most useful for each individual patient.



Medical and nursing staff will also be involved in diagnosing the causes for malnutrition. Social workers, psychologists and community health teams may also have a role to play in individual cases.

Uraemic symptoms will need to be assessed to ensure that they are adequately controlled. If they are not, the reasons for this will need to be determined and rectified¹⁹⁻²².

Other systemic diseases, especially inflammatory conditions (infection, non-functioning transplants, etc.) and intestinal disease will need to be diagnosed and treated where possible²³.

Dentition may need some attention to make it easier to chew food and address caries & gum disease.

Depression reduces food intake so this may require support or treatment if present²⁴.

Changes in social circumstances may influence the availability of food.

Some medication can influence appetite and could be reviewed.

Nutritional status is known to deteriorate as chronic disease progresses²⁵, and is a strong predictor for increased morbidity and mortality. Assessment of nutritional status should therefore be considered when patients begin education for kidney replacement treatment as part of their overall care as well as for potential intervention regarding salt, potassium, phosphate and protein / energy intake assessments¹⁵. Dietetic assessment is needed at dialysis commencement ^{8, 26} and if the mode of dialysis changes. This is an important time to assess nutritional status and dietary knowledge in terms of potassium, phosphate, salt and fluid. It is also a good time to reassess the patient's individualized nutritional care plan, prioritising which bits of information are most important for each person. More than one visit is likely to be needed. The advice given may change over time depending upon the response to dialysis and other changing circumstances.

For stable patients, nutritional changes are likely to be gradual after this¹⁹. Screening should help to remind both front line staff and patients that nutritional status can change quickly. If there are concerns about weight loss, changes in physical appearance or reported reduced food intake between screening cycles, screening should be repeated earlier or advice sought from specialist renal dietitians. The importance of multidisciplinary working between doctors, nurses and specialist renal dietitians in this regard cannot be over-emphasised.

Guideline 1.3 - We recommend that in-patients at risk of malnutrition on screening are also considered at risk of refeeding syndrome. (1D)

Rationale

The criteria used to identify those at risk in accordance with NICE guideline CG 32⁹ can be located, along with the management of refeeding syndrome on the British Association of Parenteral and Enteral Nutrition website on the following link:

www.bapen.org.uk/nutrition-support/assessment-and-planning/nutritional-assessment?start=2



Refeeding problems include life threatening acute micronutrient deficiencies, fluid and electrolyte imbalances and disturbances of metabolic processes which result from over-rapid or unbalanced nutrition support²⁷. Patients at risk of refeeding syndrome require prompt assessment by dietitians and careful multidisciplinary working.

Biochemical changes that can occur as a result of refeeding include hypophosphataemia, hypokalaemia, hypomagnesaemia and occasionally hypocalcaemia. Acute clinical conditions can result from this including cardiac failure, pulmonary oedema and dysrhythmia, acute circulatory fluid overload or fluid depletion and deterioration in neurological state and confusion. When assessing biochemical markers, use pre-dialysis bloods to avoid over-supplementation.

Thiamine is an essential coenzyme for carbohydrate metabolism, if depleted due to inadequate intakes/starvation this can lead to Wernicke's encephalopathy or Korsakoff's syndrome. Replacement of thiamine and other B vitamins is an important component of managing risk of refeeding syndrome²⁶. The link above also gives recommendations for vitamin replacement regimes.

References

- Carrero JJ,Thomas F, Nagy, Arogundade K, Avesani CM,Chan M, Chmielewski M, Cordeiro AC, Espinosa-Cuevas A, Fiaccadori E, Guebre-Egziabher F, Hand RK, Hung AM, Ikizler TA, Johansson LR, Kalantar-Zadeh K, Karupaiah T, Lindholm B, .Kovesdy CP. Global prevalence of proteinenergy wasting in kidney disease: a meta-analysis of contemporary observational studies from the international society of renal nutrition and metabolism. Journal of Renal Nutrition 2018; 28; 380-392
- **2.** Elia M, Russell CA, Stratton RJ. Malnutrition in the UK: policies to address the problem. Proceedings of the Nutrition Society. 2010; 69: 470-476
- **3.** Guest JF, Panca M, Baeyens JP, de Man F, Ljungqvist O, Pichard C, Wait S, Wilson L. Health economic impact of managing patients following a community-based diagnosis of malnutrition in the UK.. Clinical Nutrition 2011; 30: 422-429
- **4.** National Institute of Health and Care Excellence. Nutrition support for adults: oral nutrition support, enteral tube feeding and parenteral nutrition.2006; nice.org.uk/guidance/cg32
- 5. National Institue of Health and Care Excellence. Nutrition support in adults. 2012; nice.org.uk/guidance/qs24
- Omidvari AH, Vali Y, Murray SM, Wonderling D, Rashidian A. Nutritional screening for improving professional practice for patient outcomes in hospital and primary care settings. Cochrane Database of Systematic Reviews 2013, Issue 6. Art. No.: CD005539. DOI: 10.1002/14651858.CD005539.pub2.



- Palmer SC, Maggo JK, Campbell KL, Craig JC, Johnson DW, Sutanto B, Ruospo M, Tong A, Strippoli GFM. Dietary interventions for adults with chronic kidney disease. Cochrane Database of Systematic Reviews 2017, Issue 4. Art. No.: CD011998. DOI: 10.1002/14651858.CD011998.pub2.
- 8. National Institue of Health and Care Excellence. Renal replacement therapy and conservative management. 2018; nice.org.uk/guidance/ng107
- Lawson CS, Campbell KL, Dimakopoulos I, Dockrell ME. Assessing the validity and reliability of the MUST and MST nutrition screening tools in renal inpatients. Journal of Renal Nutrition. 2012;22(5):499-506
- **10.** Yamada K, Furuya R, Takita T, Maruyama Y, Yamaguchi Y, Ohkawa S, Kumagai H. Simplified nutritional screening tools for patients on maintenance haemodialysis. American Journal of Clinical Nutrition. 2008; 87: 106-113
- Role of nutrition impact symptoms in predicting nutritional status and clinical outcome in hemodialysis patients: a potential screening tool. Campbell KL, Bauer JD, Ikehiro A, Johnson DW. Journal of Renal Nutrition. 2013;23:302-7
- Xia YA, Healy A, Kruger R. Developing and validating a renal nutrition screening tool to effectively identify undernutrition risk among renal inpatients. Journal of Renal Nutrition. 2016; 26: 299-307.
- Jackson HS, MacLaughlin HL, Vidal-Diez A, Banerjee D. A new in patient renal nutrition screening tool (renal iNUT): a multicentre validation study. Clinical Nutrition 2018; http://dx.doi.org/10.1016/j.clnu.2018.10.002
- MacLaughlin HL, Twomey J, Saunt R, Blain S, Campbell KL, Emery P. The nutrition impact symptoms (NIS) score detects malnutrition risk in patients admitted to nephrology wards. J Human Nutr Diet 2018. 31: 683-688.
- **15.** National Institute of Health and Care Excellence. Chronic kidney disease in adults: assessment and management. 2014. Nice.org.uk/guidance/cg182
- 16. White JV, Guenter P, Jensen G, Malone A, Schofield M. Consensus statement: Academy of Nutrition and Dietetics and American Society for Parenteral and Enteral Nutrition: Characteristics recommended for the identification and documentation of adult malnutrition (undernutrition). Journal of Parenteral and Enteral Nutrition 2012; 36, 275-283



- **17.** Enia G, Sicuso C, Alati G, Zoccali C. Subjective global assessment of nutrition in dialysis patients. Nephrology Dialysis Transplantation 1993;8:1094-1098
- Changes in body composition in the two years after initiation of haemodialysis: a retrospective cohort study. Keane D, Gardiner C, Lindley E, Lines S, Woodrow G, Wright M. Nutrients 2016; 8:702-711
- Lindsay RM, Spanner E. A hypothesis: the protein catabolic rate is dependent upon the type and amount of treatment in dialysed uraemic patients. American Journal of Kidney Diseases 1989; 13: 382-389
- **20.** Bergstrom J, Furst P, Alvestrand A, Lindholm B. Protein and energy intake, nitrogen balance and nitrogen losses in patients treated with continuous ambulatory peritoneal dialysis. Kidney International 1993; 44: 1048-1057
- 21. Davies SJ, Phillips L, Griffiths AM, Naish PF, Russell GI. Analysis of the effects of increasing delivered dialysis treatment to malnourished peritoneal dialysis patients. Kidney International 2000; 57: 1743-1754
- 22. Rocco MV, Dwyer JT, Larive B, Greene T, Cocokram DB, Chumlea WC, Kusek JW, Leung J, Burrowes JD, McLeroy LC, Poole D, Uhlin L. The effect of dialysis dose and membrane flux on nutritional parameters in hemodialysis patients: results of the HEMO study. Kidney International 2004; 65: 2321-2334.
- **23.** Carrero JJ, Stenvinkel P, Cuppari L, Ikizler TA, Kalantar-Zadeh K, Kaysen G, Mitch W, Price R, Wanner C, Wang AYM, Wee P, Franch H. Aetiology of the protein energy wasting syndrome in chronic kidney disease: a consensus statement from the International Society of Renal Nutrition and Metabolism. Journal of Renal Nutrition 2013; 23: 77-90.
- 24. Koo JR, Park KY, Kim HJ, Che DW, Lee JS, Son BW, Rho JW. Depression in chronic haemodialysis patients: risk factors and effects on nutritional parameters. Journal of the American Society of Nephrology 1998; 9: 216A
- 25. Kopple JD, Greene T, Chumlea WC, Hollinger D, Maroni BJ, Merrill D, Scherch LK, Schulman G, Wang SR, Zimmer GS for the Modification of Diet in Renal Disease Study Group, Relationship between nutritional status and the glomerular filtration rate: results for the MDRD Study, Kidney International,2000, 20: 19-26
- 26. Fouque D, Vennegoor M, Wee PT, Wanner C, Basci A, Canaud B, Haage P, Konner K, Kooman J, Martin-Malo A, Pedrini L, Pizzarelli F, Tattersall J, Tordoir J, and Vanholder R EBPG Guideline on nutrition Nephrology Dialysis Transplantation 2007; 22 (supplement 2): ii22-ii44
- **27.** Stanga Z, Brunne A, Leuenberger M, Grimble RF, Shenkin A, Allison SP,Lobo DN Nutrition in clinical practice the refeeding syndrome: illustrative cases and guidelines for prevention and treatment. European Journal of Clinical Nutrition 2008; 62, 687–694



2. Interventions to reduce the prevalence of undernutrition in people with kidney disease (Guidelines 2.1 - 2.6)

Guideline 2.1 – Dose of small solute removal to prevent anorexia

We recommend that dialysis dose meets recommended solute clearance index guidelines (e.g. URR, Kt/V).(1C)

Rationale

Dialysis treatment to current Kt/V or URR standards is associated with better nutrient intake than lower doses^{1,2}. Attempts to increase the small solute clearance further have not demonstrated progressive improvement^{3,4}.

Current recommendations can be accessed here: https://renal.org/guidelines/

Guideline 2.2 – Correction of metabolic acidosis

We suggest that venous bicarbonate concentrations should be maintained in the normal range (2C)

Rationale

Acidosis is an established catabolic factor⁵ and the bicarbonate concentration of patients on peritoneal dialysis and HD patients should be maintained within target range to minimise this⁶. Recent data suggests that predialysis serum bicarbonate levels of >28mmol/l may be associated with adverse outcomes⁷. Bicarbonate supplementation in the low clearance clinic may retard the progression of kidney disease⁸.

Guideline 2.3 – Daily dietary protein intake

We recommend a minimum protein intake of:

- 0.8-1.0 g/kg ideal body weight (IBW)/day for patients with stage 4-5 CKD not on dialysis (1C)
- 1.1 g/kg IBW/day for patients treated with haemodialysis (1C)
- 1.0-1.2 g/kg IBW/day for patients treated with peritoneal dialysis (1C)
- This should be accompanied by an adequate energy intake. (1C)
- We do not believe there is sufficient evidence to recommend low protein diets for people with progressive kidney failure (1C)

Rationale

Before dialysis was widely available, advanced kidney disease was treated by greatly restricting protein intake. This reduced the generation of uraemic toxins. In theory, it could slow the rate of



damage to kidneys as well⁹. There has been interest in using this approach to delay the point at which dialysis begins. Conversely, there has also been concern that such an approach is likely to lead to malnutrition with the associated loss of function and poor quality of life. A Cochrane review looked at 17 studies including 2996 adults without diabetes¹⁰. They concluded that very low protein diets (0.3-0.4 g/kg/day) may delay the point at which dialysis is needed. More modest protein restriction (0.5-0.6 g/kg/day) does not have an effect. They observed that there wasn't enough information available to confirm that quality of life is maintained on very low protein diets. Concordance with these diets was difficult.

A separate Cochrane review from 2007 looked at protein restriction in diabetic adults¹¹. This did not find sufficient evidence to recommend protein restriction, though they noted that the low protein diet groups had not restricted their intake to the prescribed level in the study protocols.

The recommended protein intakes for dialysis patients are derived from a literature review carried out in 2013¹² and a systematic review published in 2014¹³. Both recognise that the evidence is not strong. Protein intakes less than 0.8g/kg and more than 1.4g/kg have been associated with increased risk of mortality¹²,¹⁴. In the presence of inter-current illness protein requirements may be higher. Recent recommendations for patients with acute kidney injury and intercurrent illness can be accessed at:

https://www.thinkkidneys.nhs.uk/aki/wp-content/uploads/sites/2/2017/12/Think-Kidneys-Nutrition-Guide-final.pdf

An adequate energy intake is needed to ensure that consumed protein is used effectively by the body¹⁵.

Guideline 2.4 – Daily energy intake

We suggest a prescribed energy intake of

- 30-40 kcal/kg IBW/day for all patients depending upon age and physical activity (2C).
- We note that peritoneal dialysis patients are likely to absorb glucose from their dialysis fluid and this should be taken into account.

Rationale

The recommended energy intake is consistent with other guidelines¹⁶⁻¹⁸. As above, it is important that this energy intake is achieved to ensure that dietary protein is used effectively. We have not identified any new evidence that this should change. An individual's recommended intake will depend upon their age, gender, level of physical activity and intercurrent medical issues. Peritoneal dialysis patients absorb some glucose from their dialysis fluid. This will vary depending upon their dialysis regime and solute transporter status. The amount of absorption can be quantified by measuring effluent volume and glucose concentration and comparing this with the total glucose content of the prescribed fluid regime that day¹⁹.



Guideline 2.5 – Micronutrient supplementation in patients on dialysis

We suggest that water soluble vitamin supplements should be offered to dialysis patients with a reduced nutrient intake or those that have unusually high levels of solute clearance on dialysis (e.g. daily or overnight haemodialysis). (2C)

We recommend that other micronutrients are supplemented only if there are symptoms consistent with deficiency and biochemical evidence of deficiency. (1C)

Rationale

Haemodialysis removes water soluble vitamins from the circulation. Data from the DOPPS database suggested that supplementation of water soluble vitamins was associated with significantly lower mortality rates at patient and institution level²⁰. A subsequent systematic review gathered information on vitamin levels in (mostly) American patients on haemodialysis²¹. They found that there was some evidence of low thiamine levels in some studies and vitamin B6 levels can also be low. Other water soluble vitamins were usually present in normal levels or there was no data. The authors caution that there may be differences between nations given that different countries fortify foods in different ways. Most of the source data was old. Dialysis patterns have changed during the intervening years. It is possible that there is greater removal of vitamins with modern treatment regimes.

A recent review gathered more up to date information on dialytic removal of micronutrients in haemodialysis²² and peritoneal dialysis²³. A very recent case study demonstrated that depletion of water soluble vitamins is a real risk in this group of people, although it is rare²⁴.

We suggest that supplementation should not be offered routinely, but it should be considered if nutritional assessment suggests that dietary intake of vitamins is likely to be low or dialysis dose is unusually high. The European Best Practice Guidelines cite recommended intake levels for the various water soluble vitamins¹⁶. There is no reason to routinely supplement trace elements such as zinc or selenium²⁵. Supplements should be considered if there are symptoms consistent with deficiency and low blood levels.

References

- Lindsay RM, Spanner E. A hypothesis: the protein catabolic rate is dependent upon the type and amount of treatment in dialysed uraemic patients. American Journal of Kidney Diseases 1989; 13: 382-389
- 2. Bergstrom J, Furst P, Alvestrand A, Lindholm B. Protein and energy intake, nitrogen balance and nitrogen losses in patients treated with continuous ambulatory peritoneal dialysis. Kidney International 1993; 44: 1048-1057



- Davies SJ, Phillips L, Griffiths AM, Naish PF, Russell GI. Analysis of the effects of increasing delivered dialysis treatment to malnourished peritoneal dialysis patients. Kidney International 2000; 57: 1743-1754
- 4. Rocco MV, Dwyer JT, Larive B, Greene T, Cocokram DB, Chumlea WC, Kusek JW, Leung J, Burrowes JD, McLeroy LC, Poole D, Uhlin L. The effect of dialysis dose and membrane flux on nutritional parameters in hemodialysis patients: results of the HEMO study. Kidney International 2004; 65: 2321-2334
- Garibotto G, Russo R, Sofia A, Sala MR, Robaudo C, Moscatelli P, Deferrari G, Tizianello A. Skeletal muscle protein synthesis and degradation in patients with chronic renal failure. Kidney Int 1994;45:1432-1439
- 6. Movilli E, Zani R, Carli O, Sangalli L, Pola A, Camerini C, Cancarini GC, Scolari F, Feller P, Maiorca R. Correction of metabolic acidosis increases serum albumin concentrations and decreases kinetically evaluated protein intake in haemodialysis patients: a prospective study. Nephrol Dial Transplant 1998;13:1719-1722
- Basile C, Rossi L, Lomonte C. Dialysate bicarbonate concentration: Too much of a good thing? Seminars in Dialysis 2018; 31: 576-582
- 8. De Brito-Ashurst I, Varagunam M, Raftery MJ, Yaqoob MM. Bicarbonate supplementation slows progression of CKD and improves nutritional status. Journal of the American Society of Nephrology 2009; doi: 10.1681/ASN.2008111205
- **9.** Kalantar-Zadeh K, Fouqe D. Nutritional management of chronic kidney disease. New England Journal of Medicine 2017; 377: 1765-1776
- 10.Hahn D, Hodson EM, Fouque D. Low protein diets for non-diabetic adults with chronic kidney disease. The Cochrane Database of Systematic Reviews. 2018; http://dx.doi.org/10.1002/14651858.CD001892.pub4
- **11.**Robertson L, Waugh N, Robertson A. Protein restriction for diabetic renal disease. The Cochrane Database of Systematic Reviews 2007; http://dx.doi.org/10.1002/14651858.CD002181.pub2
- 12.Naylor HL, Jackson H, Walker GH, Macafee S, Magee K, Hooper L, Stewart L, MacLaughlin HL. British Dietetic Association evidence-based guidelines for the protein requirements of adults undergoing maintenance haemodialysis or peritoneal dialysis. Journal of Human Nutrition and Dietetics2013; 26:315-328
- **13.**Ash S, Campbell KL, Bogard J, Millichamp A. Nutrition prescription to achieve positive outcomes in chronic kidney disease: a systematic review. Nutrients 2014; 6: 416-451



- 14.Shinaberger CS, Kilpatrick RD, Regidor DL, McAllister CJ, Greenland S, Kopple JD, Kalantar-ZadehK. Longitudinal associations between dietary protein intake and survival in hemodialysis patients.American Journal of Kidney Disease. 2006; 48:37-49
- **15.**Slomowitz LA, Monteon FJ, Grosvenor M, Laidlaw S, Kopple JD, Effect of energy intake on nutritional status in maintenance haemodialysis patients. Kidney International 1989; 35: 704-711
- **16.**Fouque D, Vennegoor M, Wee PT, Wanner C, Basci A, Canaud B, Haage P, K**o**nner K, Kooman J, Martin-Malo A, Pedrini L, Pizzarelli F, Tattersall J, Tordoir J, and Vanholder R. EBPG Guideline on nutrition Nephrology Dialysis Transplantation 2007; 22 (supplement 2): ii22-ii44
- **17.**Cano N, Fiaccadori E, Tesinsky P, Toigo G, Druml W, Kuhlmann M, Mann H, Horl WH, ESPEN guidelines on enteral nutrition: adult renal failure. Clin Nutr 2006; 25, 295-310.
- 18. Ikizler TA, Cano N, Franch H, Fouque D, Himmelfarb J, Kalantar-Zadeh K, Kuhlmann MK, Stenvinkel P, TerWee P, Teta D, Wang AYM, Wanner C. Prevention and treatment of protein energy wasting in chronic kidney disease patients: a consensus statement by the International Society of Renal Nutrition and Metabolism. Kidney Int 2013; 84, 1096-1107
- **19.** Daugirdas JT, Blake PG, Ing TS. Handbook of dialysis Third edition; 2001; 281
- 20.Fissell RB, Bragg-Gresham JL, Gillespie BW, Goodkin DA, Bommer J, Saito A, Akiba T, Port FK, Young EW. International variation in vitamin prescription and association with mortality in the Dialysis Outcomes and Practice Patterns Study (DOPPS). American Journal of Kidney Disease 2004; 44(2):293-299
- **21.**Tucker BM, Safadi S, Friedman AN. Is routine multivitamin supplementation necessary in US chronic adult haemodialysis patients? A systematic review. Journal of Renal Nutrition 2015; 25: 257-264.
- **22.**Jankowska M, Rutkowski B, Debska-Slizien A. Vitamins and Microelement Bioavailability in Different Stages of Chronic Kidney Disease. Nutrients 2017; 9: 282
- **23.**Jankowska M, Lichodziejewska-Niemierko M, Rutkowski B, Debska-Slizien A, Malgorzewic S. Water soluble vitamins and peritoneal dialysis State of the art. Clinical Nutrition 2016; 36: 1483-1489.
- **24.**Panchal S, Schneider C, Malhotra K. Scurvy in a haemodialysis patient. Rare or ignored? Haemodialysis International 2018. https://doi.org/10.1111/hdi.12705
- **25.**Tonelli M, Wiebe N, Thompson S, Kinniburgh D, Klarenbach SW, Walsh M, Bello AK, Faruque L, Field C, Manns B, Hemmelgarn BR. Trace element supplementation in haemodialysis patients: a randomized controlled trial. BMC Nephrology 2015; 16: 52



3. Interventions to treat undernutrition in people with kidney disease (Guidelines 3.1 - 3.4)

Guideline 3.1 - Anabolic agents in kidney disease

We recommend that anabolic agents should not be used to treat undernutrition in people with kidney disease. (1C)

Rationale

These have been used in small studies. A large study using Growth Hormone terminated early. It did not demonstrate benefit in mortality or quality of life¹. Anabolic steroids have significant serious side effects which greatly limit their potential for use²

Guideline 3.2 Oral nutritional supplements in patients who are undernourished

We recommend the use of oral nutritional supplements (ONS) when nutritional intake fails to increase, despite intervention and advice, and remains inadequate to meet energy and protein requirements. (1C)

Rationale

Nutrition support should be implemented to reverse diagnosed malnutrition and should involve 'food first' by the means of high energy high protein snacks, little and often approach to eating and food fortification³ The nutritional intake of patients on haemodialysis was improved with intervention from a specialist renal dietitian⁴, as was the patients' quality of life⁵. A systematic review from 2005 included eighteen studies, of which five were randomised controlled trials⁶. It found that ONS improved serum albumin and dietary intakes but there was inadequate evidence to show improved outcomes such as quality of life and mortality. Oral nutritional supplements have been found to reduce hospitalisation⁷. A more recent systematic review also found oral nutritional support improved some nutritional markers⁸

There have been recent publications to support the routine prescription of ONS treatment in patients on haemodialysis ⁷⁻¹⁰ and peritoneal dialysis¹¹. They imply that patients engaging with dietetic support programs do better than those that don't. There have been several studies looking at intradialytic oral nutrition, whereby nutritional supplements are consumed on haemodialysis. This method of nutrition support has been associated with reduced hospitalisation rates and mortality risk^{30, 12, 13}.

Guideline 3.3 - Enteral feeding in patients who are undernourished

We suggest that the use of enteral feeding is considered in selected cases if nutrient intake is suboptimal despite ONS recognising that there are significant risks and inconvenience associated with these forms of feeding (2C). It is important to also consider the patient's comorbidity, general condition and likely survival prospects before initiating enteral feeding.



Rationale

The evidence to support enteral tube feeding is limited to a few small observational studies^{14, 15}. A review of 181,196 percutaneous endoscopic gastrostomy (PEG) insertion procedures in the US found that patients with CKD had a 1.6-fold increased risk of mortality¹⁶. Having malnutrition increased mortality risk to 5.25-fold when compared to patients with head and neck disease preventing oral intake. It is important to note that the review did not state the stage of kidney disease, if the patients were receiving renal replacement therapy or how they defined malnutrition.

Guideline 3.4 - Parenteral nutritional support in patients who are undernourished

We suggest intradialytic parenteral nutrition (IDPN) in haemodialysis or intraperitoneal amino acids in peritoneal dialysis may be considered for selected cases when oral or enteral intake is suboptimal (2D).

Intradialytic parenteral nutrition (IDPN) can be used as a form of nutrition support in maintenance patients on haemodialysis. The FINE study¹⁷ compared ONS to ONS and IDPN in a randomised controlled trial. Survival rates did not differ between the two groups, and both groups improved nutritional markers. The study did not report data on quality of life, hospital admission rates or length of stay. A subsequent systematic review set out to identify if IDPN improved survival, quality of life, or nutritional status¹⁸. It concluded that there was insufficient evidence to demonstrate benefit or harm with the use of IDPN in malnourished haemodialysis patients. A more recent randomised controlled trial found a change in a biochemical marker but no improvement in quality of life scores or SGA rating¹⁹. Despite the lack of evidence for widespread use of IDPN, it may be helpful in a small number of selected cases when other options have been exhausted.

Use of "total" parenteral nutrition is recommended as per BAPEN/ESPEN guidelines when the gut is non-functioning or inaccessible.

Intraperitoneal amino acid (IPAA) solutions use 1.1% amino acid-based solution in place of a dextrose-based peritoneal dialysate. Studies that have shown IPAA with bicarbonate buffered fluid increase protein synthesis²⁰. It appears that having the amino acid solution administered in a fed state helps to increase protein synthesis²¹. The studies were short in duration and failed to see a significant improvement in nitrogen balance. In a 3-year, randomized controlled study, 60 malnourished CAPD patients were randomly assigned to either replace one daily exchange with IPAA or to continue with dextrose dialysate²². Dietary protein intake increased in the IPAA group. Biochemical nutritional parameters stayed stable or increased in the IPAA group but decreased in the dextrose group. There was no effect on patient survival. As such, this product may have a place in the treatment of a small number of selected individuals.



References

- Kopple JD, Cheung AK, Christiansen JS, Djurhuus CB, El Nahas M, Felt-Rasmussen B, Mitch WE, Wanner C, Gothberg, M Ikizler TA. OPPORTUNITY: a large-scale randomized clinical trial of growth hormone in hemodialysis patients. Nephrology Dialysis Transplantation 2011; 26: 4095-4103
- **2.** Orr R, Fiatarone Singh M. The anabolic androgenic steroid oxandrolone in the treatment of wasting and catabolic disorders: review of efficacy and safety. Drugs; 2004: 64: 725-750
- **3.** Stenvinkel, P. Inflammatory and atherosclerotic interactions in the depleted uremic patient. Blood Purification 2001; 19: 53-61
- Campbell KL, Ash S, Zabel R, McFarlane C, Juffs P, Bauer JD. Standardised nutrition guidelines by renal dietitians is associated with improved nutritional status. Journal of Renal Nutrition 2009, 19: 136-144
- **5.** Campbell KL, Ash S, Bauer JD. The impact of nutrition intervention on quality of life in pre-dialysis chronic kidney disease patients. Clinical Nutrition 2008; 27: 537-544
- Stratton RJ, Bircher G, Fouque D, Stenvinkel P, de Mutsert R, Engfer M, Elia M. Multinutrient oral supplements and tube feeding in maintenance dialysis: a systematic review and meta-analysis. American Journal of Kidney Disease 2005, 46, 387-405
- Cheu C, Pearson J, Dahlerus C, Lantz B, Chowdhury T, Sauer PF, Farrell RE, Port FK, Ramirez SP. Association between oral nutritional supplementation and clinical outcomes among patients with ESRD. Clinical Journal of the American Society of Nephrology 2012; 8: 100-107
- Liu PJ., Ma F, Wang QY, He SL. The effects of oral nutritional supplements in patients with maintenance dialysis therapy: A systematic review and meta-analysis of randomized clinical trials. PLOS ONE 2018; https://doi.org/10.137 1/journal.pone.0203706
- **9.** Sezer S, Bal Z, Tutal E, Uyar ME, Acar NO. Long-Term oral nutrition supplementation improves outcomes in malnourished patients with chronic kidney disease on hemodialysis. Journal of Parenteral and Enteral Nutrition 2014; 38: 960-965
- **10.**Lacson E, Wang W, Zebrowski B, Wingard R, Hakim RM. Outcomes associated with intradialytic oral nutritional supplements in patients undergoing maintenance haemodialysis: a quality improvement report. American Journal of Kidney Disease 2012; 60: 591-600.
- 11.Satirapoj B, Limwannata P, Kleebchaiyaphum C, Prapakorn J, Yatinan U, Chotsriluecha S, Supasyndh O. Nutritional status among peritoneal dialysis patients after oral supplement with ONCE dialyze formula. International Journal of Nephrology and Renovascular Disease 2017; 10:145-151



- **12.**Calgar, K. Fedje, L. Dimmitt, R. Hakim, R.M. Shyr, Y, Ikizler TAJ. Therapeutic effects of oral nutritional supplementation during hemodialysis. Kidney International 2002; 62: 1054-1059
- **13.**Pupim LB, Majchrzak KM, Flakoll PJ, Ikizler TAJ. Intradialytic oral nutrition improves protein homeostasis in chronic hemodialysis patients with deranged nutritional status. Journal of the American Society of Nephrology. 2006 17: 3149-57.
- **14.**Sayce H, Rowe P, McGonogle R. Percutaneous endoscopic gastrostomy feeding in haemodialysis out-patients. Journal of. Human Nutrition and Dietetics 2000; 13: 333-341
- **15.**Holley J, Kirk J. Enteral tube feeding in a cohort of chronic haemodialysis patients. Journal of Renal Nutrition 2002; 12: 177-182
- **16.**Arora G, Rockey D, Gupta S. High In-hospital mortality after percutaneous endoscopic gastrostomy: results of a nationwide population-based study. Clinical Gastroenterology and Hepatology 2013; 11: 1437-1444
- 17.Cano NJ, Fouque D, Roth H, Aparicio M; Azar R; Canaud B; Chauveau P; Combe C; Laville M; Leverve XM. Intradialytic parenteral nutrition does not improve survival in malnourished haemodialysis patients: a 2 year multi-centre prospective, randomized study. Journal of the American Society of Nephrology 2007; 18: 2583-259
- **18.**Sigrist MK, Levin A, Tejani AM Systematic review of evidence for the use of intradialytic parenteral nutrition in malnourished hemodialysis patients. Journal of Renal Nutrition. 2010 ;20:1-7
- **19.**Marsen TA; Beer J; Mann H. Intradialytic parenteral nutrition in maintenance hemodialysis patients suffering from protein-energy wasting. Results of a multicenter, open, prospective, randomized trial. Clinical Nutrition 2017;36:107–117
- **20.**Tjiong HL, van den Berg JW, Wattimena JL, et al. Dialysate as food: combined amino acid and glucose dialysate improves protein anabolism in renal failure patients on automated peritoneal dialysis. Journal of the American Society of Nephrology 2005;16:1486-1493.
- **21.**Tjiong HL, Rietveld T, Wattimena JL, van den Berg JW, Kahriman D, van der Steen J, Hop WC, Swart R, Fieren MW. Peritoneal dialysis with solutions containing amino acids plus glucose promotes protein synthesis during oral feeding. Clinical Journal of the American Society of Nephrology. 2007; 2:74-80.
- **22.** Li FK, Chan LY, Woo JC, et al. A 3-year, prospective, randomized, controlled study on amino acid dialysate in patients on CAPD. American Journal of Kidney Disease 2003; 42:173-183.



4. Lay Summary

Undernutrition is common in people with kidney disease. It is linked to problems that can affect quality of life and wellbeing. Based on the research that has been done, the risk of not being able to eat enough food to maintain usual weight is higher in people with kidney disease than in those without kidney disease.

We recommend that all people with advanced kidney disease have access to a specialist renal dietitian to help understand more about how to eat healthily. Diet affects several aspects of kidney disease and it can be tricky to find a balance between restricting things that can cause problems and taking enough of the things that are needed.

Kidney services should be able to look out for signs that undernutrition is developing and have a detailed plan about what to do should it be identified. If it looks as though this is happening, those affected should be given the opportunity to talk to a dietitian who is qualified to treat undernutrition in people with kidney disease.

Sometimes, there can be more than one reason for undernutrition, so several different specialists might need to work together to help. When someone cannot eat enough ordinary food to stay well nourished, supplements or other forms of additional nutrition may be recommended.

It appears that regular exercise is good for people having dialysis. It can help to preserve muscle function and makes people feel better.